Author’s response to reviews

Title: Surgical Correction of Hyperlordosis in Facioscapulohumeral Muscular Dystrophy. A Case Report

Authors:

Haining Tan (tanhaining2@163.com)
Fan Feng (drfengfan@163.com)
Youxi Lin (linuc5@126.com)
Chong Chen (markchan720@163.com)
Zheng Li (kleeo@163.com)
Jianxiong Shen (shenjianxiong@medmail.com.cn)

Version: 2 Date: 28 Jun 2017

Author’s response to reviews:

Dear Prof. Helen,

Thank you very much for the comprehensive review of our manuscript entitled: "Surgical Correction of Hyperlordosis in Facioscapulohumeral Muscular Dystrophy. A Case Report" (BSUR-D-17-00076). We sincerely appreciate the editorial requests and provide a point-by-point response to the requests.

(1) Request: Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them. Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them.

Response: The “Figure Legends” is added as a separate section after the References. (page 7)

(2) Request: Please amend your references to comply with our submission guidelines: All references, including URLs, must be numbered consecutively, in square brackets, in the order in which they are cited in the text, followed by any in tables or legends. The reference numbers must be finalized and the reference list fully formatted before submission.

Response: All references are amended to comply with the submission guidelines. (page 6)
(3) Request: Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication.

Response: Our manuscript is edited by a native English speaker, and some corrections are adopted in this revision. If there still are some issues needed to be addressed, please notice us.

Many thanks to you, Prof. Helen, for the time and efforts spent on this paper. I sincerely hope this revised manuscript will be accepted for publication.